Author's response to reviews

Title: beta2-microglobulin induce epithelial-mesenchymal transition in human renal proximal tubule epithelial cells in vitro

Authors:

Ai qing Zhang (njaqing@njmu.edu.cn)
Bin Wang (wangbinhewei@126.com)
Min Yang (yangmin79@sina.com)
Hui min Shi (pjr1951@163.com)
Wei hua Gan (weihuagan@njmu.edu.cn)

Version: 5 Date: 26 February 2015

Author's response to reviews:

Editorial Request:
1. In the abstract change Objective to Background
Answer: Done accordingly
2. Please remove the copies of letter to reviewers or editors from the additional files.
Answer: Done accordingly